Author’s response to reviews

Title: Medical Students’ Perception on Fecal Microbiota Transplantation

Authors:

Petru Madar (petrumadar@gmail.com)
Oana Petre (oanapetre@psychology.ro)
Adriana Baban (adrianababan@psychology.ro)
Dan Dumitrascu (dumitrascu@umfcluj.ro)

Version: 3 Date: 05 Sep 2019

Author’s response to reviews:

Dear Editor,

We have addressed the issues pointed out in the last decision letter.